Author's response to reviews

Title: Effectiveness of a complex intervention on reducing the prevalence of smoking among adolescents: study design of a cluster-randomised, controlled trial.

Authors:

Alfonso Leiva (aleiva@ibsalut.caib.es)
Andreu Estela (andreu.estela@ssib.es)
Maties Torrent (maties.torrent@ssib.es)
Amador Calafat (irefrea@irefrea.org)
Miquel Bennasar (miquel.bennasar@uib.es)
Aina Yáñez (ainam.yanez@ssib.es)

Version: 3  
Date: 25 February 2014

Author's response to reviews:

Dear Editor

The manuscript was edited, I feel this time the language improve.


We apologize, but we also has found two missing categories in the smoking questionnaire, D) I currently smoke at least one cigarette per month, but less than one cigarette per week and G) I used to smoke regularly in the past, but I do not smoke now.

It is also in the methods section describe the administrative organization of the Balearic island:” The Balearic island is organized into four insular councils (Majorca, Minorca, Ibiza and Formentera) and those in municipalities. There is 53 municipalities in Majorca and 8 in Minorca, between 300 to 350,000 inhabitants. “

Yours sincerely